UNIFORM NETWORK CODE - GENERAL TERMS

SECTION D - CDSP AND UK LINK

1 GENERAL

1.1 Introduction

- 1.1.1 This Section D sets out:
 - (a) provisions relating to the CDSP and CDSP Services;
 - (b) requirements in respect of the Data Services Contract;
 - (c) the basis on which the DSC Committees are established; and
 - (d) provisions in relation to UK Link.
- 1.1.2 This Section D includes the provisions required to be included in the Code pursuant to the CDSP Licence Condition.
- 1.1.3 The provisions of this Section D apply subject to the Transition Document (including the Transitional Arrangements Document referred to in the Transition Document).

1.2 Interpretation

- 1.2.1 For the purposes of the Code:
 - the "Central Data Services Provider" or "CDSP" is the person for the time being appointed by the Transporters as central data services provider pursuant to the CDSP Licence Condition;
 - (b) the "CDSP Licence Condition" is Standard Special Condition A15 of Gas Transporter's Licences;
 - (c) the "Data Services Contract" or "DSC" is the contract between the Parties and the CDSP in the agreed form (as provided in Part I paragraph 3 of the Transition Document) as from time to time amended in accordance with its terms and the provisions of this Section D;
 - (d) **"Core Customer"** means a Party to the Code, other than a Trader User, in the capacity of a party to the DSC;
 - (e) "Customer Class" means each of the following classes of Core Customer:
 - (i) Shipper Users;
 - (ii) DN Operators;
 - (iii) National Gas Transmission;
 - (iv) Independent Gas Transporters;

© 2005 all rights reserved

GTD 1

- (f) "Customer Group" means a single Customer Class or a group of Customer Classes;
- (g) **"CDSP Cost"** means any cost, expense, liability or other outgoing of the CDSP, however arising or incurred;
- (h) "CDSP Charge" means any charge made by the CDSP to a Core Customer or other person for the provision of a CDSP Service or otherwise for the recovery of any CDSP Costs;
- "CDSP Year" means the period from 1 April in any year until and including 31 March in the following year;
- "DCC Licence" means the 'smart meter communications licence' referred to in Section V5.17;
- (k) "IGT Code Communication" means a communication, equivalent to a TPD Communication, to be given by or to an Independent Gas Transporter under the IGT Code;
- (l) **"Retail Energy Code"** means the retail energy code referred to in the DCC Licence; and
- (m) **"GRDA Functions"** means functions assigned to the CDSP as 'Gas Retail Data Agent' in the Retail Energy Code.
- 1.2.2 The DSC Objectives are:
 - (a) compliance with the requirements of the CDSP Licence Condition, including (without limitation):
 - (i) that the CDSP should provide services effectively to help facilitate the efficient and integrated operation of the gas industry; and
 - so far as concerns CDSP Charges, compliance with the requirements (including the Charging Methodology Objectives (as defined in the CDSP Licence Condition)) in paragraph 8(d)(ii) of the CDSP Licence Condition; and
 - (b) the objectives in paragraph 1 of Standard Special Condition A11 of Gas Transporter's Licences;
 - (c) (without duplication of the objectives in paragraph (b)) facilitating effective competition between Parties or Parties of any class-; and
 - (c)(d) the costs incurred in the delivery of CDSP Services and the performance of Non-Service Functions should be efficient and economic.
- 1.2.3 Subject to paragraph 1.2.4, references in this Section D to Shipper Users include such persons in their capacity as IGTS Users.
- 1.2.4 Nothing in the Code operates to create any rights or obligations as between an Independent Gas Transporter and an IGTS User (in that capacity), such matters being the subject of the IGT Code.

1.2.5The DSC is the service agreement referred to in the CDSP Licence Condition as the© 2005 all rights reservedGTD 2Version 5.89 29 March 2023

'CDSP Service Agreement'.

© 2005 all rights reserved

1.2.6 For the purposes of paragraph 6(a) of the CDSP Licence Condition a "user of CDSP Services" is any Party other than a Trader User.

1.3 CDSP Services

- 1.3.1 For the purposes of the Code, subject to paragraphs 1.3.2 to 1.3.4, "CDSP Services" means the following services:
 - (a) services (**"core services"**) which consist in:
 - (i) performing functions assigned to the CDSP in; or
 - performing or facilitating performance of obligations on behalf of a Core Customer under

the Code, the IGT Code, the Retail Energy Code (in respect of GRDA Functions only), a Gas Transporter's Licence, or another document designated for the purposes of Section 173 of the Energy Act 2004;

- (b) any service which the CDSP can provide efficiently and economically by using resources and/or data used by the CDSP to provide core services and which:
 - (i) is provided to any Core Customer or Core Customers of any Customer Class; or
 - (ii) is provided to Trader Users; or
 - (iii) is provided to a person or persons of a class which are not Parties, where one of the following conditions is met:
 - the service is the same or substantially the same as a service which the CDSP is for the time being providing to any Core Customer or Core Customers of a Customer Class; or
 - the Code requires or authorises the CDSP to provide the service to such person(s); or
 - (3) the provision of the service to such person(s) helps facilitate the efficient and integrated operation of the gas industry; and
- (c) services which consist of, are connected with or are ancillary to the CDSP performing the role, functions and responsibilities of the EPG Scheme Administrator (and acting as the agent of the Secretary of State in respect of such role, functions and responsibilities) in connection with the Code and the Data Services Contract ("EPG Scheme Services").
- 1.3.2 Paragraph 1.3.1 is without prejudice to the provisions of this Section D and the DSC which determine which CDSP Services the CDSP actually provides.
- 1.3.3 If the Authority consents (under paragraph 7(b) of the CDSP Licence Condition) to the CDSP providing a service beyond what is provided in paragraph 1.3.1, and such service is provided to Core Customer(s) under the DSC, such service shall be treated as being a CDSP Service.
- 1.3.4 If a question arises as to whether a service being or proposed to be provided by the

© 2005 all rights reserved

GTD 4

CDSP qualifies as a CDSP Service under paragraph 1.3.1(b):

- (a) the CDSP shall notify Core Customers of such question;
- (b) at the request of any Transporter the CDSP and the Transporters will consult the Authority on such question;
- (c) any Core Customer may send representations to the Transporters and the Authority in respect of such question;

and if (following such consultation) the Transporters are not satisfied that the service does so qualify, the CDSP in consultation with the Transporters shall take such steps as are necessary to cease or not to commence the provision of such service unless the Authority consents to such provision under paragraph 7(b) of the CDSP Licence Condition; and nothing done by the CDSP pursuant to this paragraph 1.3.4 shall be a breach of the DSC.

- 1.3.5 **"Code Services"** means the CDSP Services referred to in paragraphs 2.1.1(a) and 2.2.1(a) (being CDSP Services required under or in order to give effect to the Code or IGT Code).
- 1.3.6 CDSP Services are divided into:
 - (a) services provided by the CDSP on its own account ("Direct Services");
 - (b) services which comprise the performance by the CDSP of functions on behalf of one or more of the Transporters or Independent Gas Transporters ("Agency Services").

1.4 Agreement to engage the CDSP and be party to DSC

- 1.4.1 The Parties acknowledge and agree that CDSP Services which are Code Services must be performed in order to implement and give effect to the Code and the IGT Code.
- 1.4.2 Accordingly the Parties shall, and each Party undertakes that it will:
 - (a) use the Code Services;
 - (b) engage the CDSP to perform and carry out the Code Services;
 - (c) for that purpose, enter into the DSC; and
 - (d) (without limitation of its obligations under the DSC), pay CDSP Charges as applicable to each Party in accordance with the DSC and the CDSP's Annual Charging Statement.
- 1.4.3 In accordance with TPD Section V2.1.2(i), OAD Section N4.2.2(e) and IGTAD Section F3.2.2(d) no person may become a Party, other than as a Trader User, without acceding to the DSC.
- 1.4.4 Each Party undertakes that it will act and exercise its rights and powers under the DSC in relation to the CDSP so as jointly to control and govern the CDSP on an economic and efficient basis.

© 2005 all rights reserved

GTD 5

- 1.4.5
 Each Party undertakes that it will act and exercise its rights and powers under the DSC in relation to the CDSP so as jointly to ensure that costs incurred in the delivery of CDSP Services and the performance of Non-Service Functions should be efficient and economic.
- <u>1.4.51.4.6</u> A Party may appeal the CDSP Budget for a CDSP Year by notice in writing to the Authority in accordance with and subject to the process set out in the Budget and Charging Methodology provided for in paragraph 3.3.1(b).

1.4.61.4.7 The Parties agree and acknowledge that:

- (a) the CDSP will not declare dividends or distributions (of income or capital) to its shareholders;
- (b) the shareholders of the CDSP have no obligations in respect of the funding of the CDSP;
- (c) the objective of the CDSP is to operate without making a net surplus (other than a margin to provide working capital) or net loss in any CDSP Year (and, while some over or under-recovery of CDSP Costs in any one year is likely to occur, the provisions of the Budget and Charging Methodology ensure that such over or under-recovery is corrected in subsequent years); and
- (d) accordingly (in accordance with the further provisions of this Section D):
 - the CDSP Services are provided 'at cost' to Core Customers (save as to the margin referred to in paragraph (c)); and
 - all CDSP Costs (to the extent not recovered from third parties) are to be recovered from Core Customers.

1.5 Relevant Licence Provisions

- 1.5.1 For the purposes of this paragraph 1.5 "Relevant Licence Provisions" are:
 - (a) the CDSP Licence Condition;
 - (b) Standard Special Condition A31 and Standard Condition 31 of the Transporters' and Independent Gas Transporters' Gas Transporter's Licences;
 - (c) Standard Condition 14 of the Transporters' Gas Transporter's Licences, insofar as that condition applies in relation to the obligation of the Transporters under the Retail Energy Code to procure the CDSP as Gas Retail Data Agent to perform its obligations under the Retail Energy Code; and
 - (d) any other condition of the Transporters' and Independent Gas Transporters' Gas Transporter's Licences, to the extent to which the CDSP Services or other functions of the CDSP are relevant to compliance with such condition.
- 1.5.2 In the event of a conflict between a provision of this Section D or the DSC with a Relevant Licence Provision (or with compliance with a Relevant Licence Provision):
 - (a) such provision of this Section D or the DSC shall be deemed modified to the minimum extent necessary to make it consistent with the Relevant Licence Provision;

© 2005 all rights reserved

GTD 6

(b) if such modification is not possible, the relevant provision of this Section D or the DSC shall be deemed deleted; and

© 2005 all rights reserved

(c) any such modification or deletion of a provision shall not affect the validity or enforceability of any other part of this Section D or the DSC.

1.5.3 It is agreed that:

- (a) the CDSP shall not knowingly do or omit to do anything;
- (b) a DSC Committee shall not take a decision (and any such decision shall be invalid and of no effect); and
- (c) no Party shall knowingly exercise any right under this Section D or the DSC in a way,

which would result in a Transporter or Independent Gas Transporter being in breach of a Relevant Licence Provision.

1.5.4 Where a Transporter or Independent Gas Transporter considers that any possible or proposed action or decision within paragraph 1.5.3(a), (b) or (c) would result in a breach of a Relevant Licence Provision, the Transporter or Independent Gas Transporter may and shall so inform the relevant person or committee accordingly.

1.6 Trader Users

- 1.6.1 This Section D, other than paragraph 5, does not apply in relation to Trader Users (which are not users of CDSP Services for the purposes of paragraph 1.2.6) and, except in paragraph 5, references in this Section D to a Party do not include a Trader User.
- 1.6.2 Trader Users are required (pursuant to TPD Section V2.1.2(j)) to be party to a UK Link User Agreement with the CDSP.

1.7 Further provisions in relation to CDSP

- 1.7.1 The Code and IGT Code also assign certain functions (**"Non-Service Functions"**) to the CDSP which the CDSP is to perform but which are not treated or charged for as separate services provided by the CDSP, including:
 - (a) establishing and operating UK Link (except that the operation of UK Link Gemini is performed as a service to National Gas Transmission and except to any aspect of the operation of UK Link which is specified in a General Service in the CDSP Service Description);
 - (b) maintaining the UK Link Manual; and
 - (c) functions assigned under the Modification Rules in connection with the Modification Procedures,

and any function assigned to the CDSP in the Code or the IGT Code which is not specified as a CDSP Service in or pursuant to the CDSP Service Description is a Non-Service Function of the CDSP.

1.7.2 The Code also contains certain other provisions (**"CDSP-Related Provisions"**) which relate to the CDSP or the DSC as follows:

© 2005 all rights reserved

GTD 8

- (a) the provisions of paragraphs 1.3.3, 1.3.4, 1.5, 2.2.3, 2.3.1(c), 2.4.2(b), 2.4.3, 5.5.5;
- (b) any provision of the Code which is expressed to be a CDSP-Related Provision; and
- (c) any other provision of the Code (other than one which provides for a Direct Function or Non-Service Function of the CDSP) which is expressed to provide a right, entitlement, or discretion of the CDSP, or to prescribe the manner in which the CDSP is to exercise any Non-Service Function under the Code.

1.8 Functions of Code entities under DSC

1.8.1 The DSC may assign functions to the Code Administrator, Modification Panel, a Workgroup or UNC Committee, and each such person or body shall have and perform the functions so assigned to it.

1.9 EPG Scheme

- 1.9.1 For the purposes of the Code:
 - (a) "EPG Scheme" means the Energy Price Guarantee for Domestic Gas Consumers in Great Britain scheme established by the Secretary of State and set out in the EPG Scheme Document;
 - (b) "EPG Scheme Account" means the bank account established by the CDSP for the purposes of the provision of EPG Scheme Services and into which EPG Scheme Payments are received and from which ESG Scheme Payments are made;
 - (c) "EPG Scheme Document" means the document of that name issued by the Secretary of State establishing the EPG Scheme (as amended from time to time);
 - (d) "EPG Scheme Administrator" means the person designated by the Secretary of State in accordance with the EPG Scheme Document to discharge functions in connection with the operation of the EPG Scheme;
 - (e) "EPG Scheme Agreement" means the agreement between the Secretary of State, the CDSP (as the EPG Scheme Administrator) and the EPG Scheme Parties;
 - (f) **"EPG Scheme Party"** means a party to the EPG Scheme Agreement;
 - (g) **"EPG Scheme Payment"** means an amount payable by or to the Secretary of State or by or to an EPG Scheme Party in accordance with the EPG Scheme Document and the EPG Scheme Agreement.
- 1.9.2 The UNC Committee and the DSC Committees shall have no role, functions or responsibilities under the Code or the Data Services Contract in respect of:
 - (a) the implementation and operation of the EPG Scheme; or

© 2005 all rights reserved

GTD 9

- (b) the performance by the CDSP of EPG Scheme Services.
- 1.9.3 Each Party, the UNC Committee and the DSC Committees shall not, whether by act or omission, hinder, restrict or otherwise restrict the CDSP in the performance of EPG Scheme Services, and each Party, the UNC Committee and the DSC Committees shall (to the fullest extent possible) co-operate with, and assist, the CDSP in the performance of EPG Scheme Services.
- 1.9.4 The CDSP as the EPG Scheme Administrator shall owe no duties or responsibilities to any Party, the UNC Committee or the DSC Committees in performing EPG Scheme Services.
- 1.9.5 The CDSP shall account for all payments to EPG Scheme Parties separately from CDSP Costs and revenues, and shall receive and hold amounts payable to and by it which comprise EPG Scheme Payments in a separate bank account and in accordance with the requirements of the EPG Scheme Document.
- 1.9.6 Each Party acknowledges that the CDSP as the EPG Scheme Administrator is not required on any day to make payment to EPG Scheme Parties of any amounts which in aggregate are greater than the amounts received from the Secretary of State and standing to the credit of the EPG Scheme Account on such day.
- 1.9.7 Each Party consents to the CDSP providing such data as is maintained by the CDSP for the purposes of the Code to such persons requiring such data where necessary for the performance by the CDSP of EPG Scheme Services and the Data Permissions Matrix may be modified by the CDSP for such purpose.
- 1.9.8 The CDSP may modify the CDSP Service Description for the purposes of performing EPG Scheme Services.
- 1.9.9 Other than as provided for in the EPG Scheme Document, each Party waives any rights or remedies, any claim in damages or any other claim of a financial nature against the CDSP, and each Party further releases the CDSP from any and all duties or liabilities arising in contract, tort (including negligence or nuisance) or otherwise (including breach of statutory duty) in relation to the CDSP's performance of EPG Scheme Services.
- 1.9.10 The costs, expenses and other liabilities incurred by the CDSP in performing EPG Scheme Services shall be CDSP Costs.
- 1.9.11 Paragraph 1.3.1(c)) and this paragraph 1.9 may not be modified without the prior written consent of the Secretary of State.
- 1.9.12 The CDSP shall cease to provide EPG Scheme Services with effect from the date on which the CDSP ceases to be the EPG Scheme Administrator.

2 CDSP SERVICES AND ROLE OF CDSP

2.1 Direct Services

- 2.1.1 Direct Services comprise:
 - (a) the performance of functions (other than Non-Service Functions) assigned to the

© 2005 all rights reserved

GTD 10

CDSP in the Code or IGT Code ("Direct Functions");

- (b) the performance of GRDA Functions;
- (c) services provided under the DSC to Core Customers which are not prescribed by the Code or IGT Code (as the case may be), comprising:
 - services provided on a uniform basis to all Core Customers or Core Customers of a Customer Class or Customer Classes;
 - services available on a uniform basis to all Core Customers or Core Customers of a Customer Class or Customer Classes, but provided only on the request of a Core Customer; and
 - services provided at the request of and as specified by an individual Core Customer ("Additional Customer Services"); and
- (d) services (**"Third Party Services"**) provided (consistent with the policy provided for in paragraph 3.7) to persons who are not Core Customers, or to Core Customers on terms other than those of the DSC.

2.1.2 Direct Services within paragraph 2.1.1(a) comprise:

- (a) performance of the following Direct Functions under the Code:
 - the maintenance of the Supply Point Register and the performance of the functions of the CDSP (in connection with supply point administration) in TPD Section G;
 - (ii) the functions of the CDSP (in connection with demand estimation) in TPD Sections H1, H2, H3 and H4;
 - (iii) the functions of the CDSP (in connection with Supply Point metering, meter data and meter reading) in TPD Section M; and
 - (iv) the Direct Functions of the CDSP under other provisions of the Code which provide for such functions; and
- (b) performance of the Direct Functions of the CDSP under provisions of the IGT Code equivalent to the provisions of the Code listed in paragraph (a) above.
- 2.1.3 The CDSP acts on its own account and not as agent of any Party in the provision of Direct Services, but without prejudice to paragraph 2.4.2.

2.2 Agency Services

- 2.2.1 Agency Services comprise:
 - the performance of certain functions of the Transporters and Independent Gas Transporters under the Code and the IGT Code, as the case may be (such functions to be performed by the CDSP being "Agency Functions");
 - (b) the performance of certain functions of the Transporters and Independent Gas Transporters pursuant to conditions of their respective Gas Transporter's

© 2005 all rights reserved

GTD 11

Licences; and

- (c) the performance of certain functions of the Transporters and/or Independent Gas Transporters in connection with other gas industry codes or agreements.
- 2.2.2 Agency Services within paragraph 2.2.1(a) comprise:
 - the performance of the Agency Functions of the CDSP in TPD Sections B, C, E, F, G, H, Q, S, V, and X and the performance of the agency functions in the equivalent provisions of the IGT Code;
 - (b) the performance of the Agency Functions of the CDSP (in connection with data flows between Transporters and Independent Gas Transporters in IGTAD Sections D and E; and
 - (c) the performance of the Agency Functions of the CDSP under other provisions of the Code which provide for such functions.
- 2.2.3 In relation to Agency Services which are Code Services, where the Agency Functions of the CDSP include sending or receiving Code Communications or making determinations on behalf of any Transporter:
 - the CDSP acts as agent of the relevant Transporter(s) in the performance of those functions;
 - (b) in accordance with paragraph (a):
 - Code Communications are given between Users and Transporters, and in sending or receiving such Code Communications the CDSP acts as agent of the Transporter or Transporters;
 - any such Code Communication given by the CDSP shall be deemed to have been given by and be binding on the Transporter, and Users shall be entitled to rely on such Code Communication; and
 - (iii) where there is a requirement in the Code under any such provision that a User give a Code Communication to a Transporter or the Transporters collectively, the User shall be treated as having complied with such requirement where the User gives the Code Communication to the CDSP; and
 - (c) where the Agency Service requires the CDSP to act on behalf of the Transporters in the exercise of any discretion or rights conferred on the Transporters, the Transporters may provide guidance or instructions to the CDSP in respect thereof.
- 2.2.4 The IGT Code contains provisions equivalent to paragraph 2.2.3.

2.3 Direct Services which are Code Services – further provisions applying in relation to the Code

2.3.1 The Parties acknowledge and agree, in connection with Direct Services which are Code Services, that:

© 2005 all rights reserved

GTD 12

- (a) the Code sets out or describes the functions of the CDSP as obligations of the CDSP (as if the CDSP were a party to the Code);
- (b) the CDSP is not a Party, and is not bound by any Framework Agreement to comply with the Code; and
- (c) accordingly the DSC operates to bind the CDSP to perform those functions assigned to it under the Code.
- 2.3.2 Where a provision of the Code referred to in paragraph 2.1.2, in relation to which the CDSP performs Direct Services, contains a rule, requirement or obligation binding on a Party and not expressed to be owed to any other Party or Parties in particular:
 - (a) such provision operates for the benefit of all Parties, notwithstanding that such provision may be expressed as a provision about a TPD Communication to be given by or to, or a process to be followed by or with, the CDSP; and
 - (b) any Party which is or would be adversely affected by a breach of such provision may seek to enforce such provision (and GT Section B 2.4.1 does not apply to this paragraph).
- 2.3.3 The rights and obligations of the Parties under the Code shall not be affected by any failure or delay on the part of the CDSP in performing the Direct Services which are Code Services, so far as such rights and obligations are capable of being construed and determined notwithstanding such failure or delay, but without prejudice to any provision of the Code which expressly addresses any such failure or delay or the consequences thereof.
- 2.3.4 In the event of any conflict between the provisions of the Code and the provisions of the DSC:
 - (a) as between the Parties, the provisions of the Code shall prevail; and
 - (b) any Party may propose (in accordance with the applicable provisions of the DSC) an amendment to the DSC to remove such conflict.
- 2.3.5 The IGT Code contains provisions equivalent to those of this paragraph 2.3.

2.4 Direct Services which are Code Services – further provisions applying in relation to the Code and IGT Code

- 2.4.1 In relation to Direct Services which are Code Services:
 - (a) the CDSP acts as the operator in its own name of the Supply Point Register and the supply point register under the IGT Code; and
 - (b) TPD Communications and IGT Code Communications given or received by the CDSP under the provisions of the Code and the IGT Code referred to in paragraph 2.1.2 are given or received by it in its own name (as such operator), and not as agent of any particular Party.
- 2.4.2 So far as any of the Direct Functions comprise the functions to be fulfilled by the Supply Point Information Service (**"SPIS"**) pursuant to Standard Special Condition

© 2005 all rights reserved

GTD 13

A31 and Standard Condition 31 ("SC31") of the Transporters' and Independent Gas Transporters' Gas Transporter's Licences:

- (a) the DSC shall take effect as an arrangement by which (as contemplated by SC31) the Transporters and the Independent Gas Transporters procure the establishment and subsequent operation and maintenance by the CDSP of the SPIS; and
- (b) the CDSP shall perform the CDSP Services and otherwise act so as to ensure the Transporters and the Independent Gas Transporters comply with SC31.
- 2.4.3 Where, in connection with Direct Services, a provision of the Code or the IGT Code confers on the CDSP any discretion or choice, or requires or entitles the CDSP to make a judgement or form an opinion, in connection with any action taken or communication given by a particular Party (the relevant Party):
 - the Contract Management Committee may prescribe principles or guidance to be followed by the CDSP in doing so;
 - (b) the CDSP may request the Contract Management Committee to provide such principles or guidance (in general or in particular cases); and
 - (c) subject to or in the absence of any such principles or guidance, the CDSP shall act with a view to meeting the following principles:
 - to avoid undue discrimination between Parties or Parties of a particular Customer Class;
 - to avoid any significant cost or risk being imposed on any Party or Parties other than the relevant Party;
 - to avoid any significant cost or risk being incurred by the CDSP (including any risk to UK Link) which may affect other Parties;
 - to allow the relevant Party to enjoy the benefit of the relevant provision where no significant cost or risk will be imposed on any other Party or Parties, but subject to paragraph (v); and
 - not to allow the relevant Party to avoid complying with the Code or IGT Code, as the case may be, on a persistent basis.
- 2.4.4 Except as expressly provided in the Code or the DSC, no provision of the Code or the IGT Code is incorporated into the DSC and nothing in the Code or the IGT Code shall operate to impose any obligation or confer any right on the CDSP.

2.5 Amendment of DSC by Code Modification

- 2.5.1 This paragraph 2.5 applies where the Code or the DSC provides that any part of the DSC may be amended, or may only be amended, where authorised by a Code Modification pursuant to the Modification Rules.
- 2.5.2 Any Party may make a Modification Proposal to authorise the amendment of the relevant part of the DSC, subject to the provisions of the DSC.

© 2005 all rights reserved

GTD 14

2.5.3 The CDSP will, in accordance with the applicable provisions of the DSC, amend the relevant part of the DSC in accordance with any Code Modification which authorises such amendment, and such amendment shall be made and effective as provided in the DSC.

2.6 Data Permissions Matrix

- 2.6.1 For the purposes of the Code:
 - (a) **"DAM"** is the 'Data Access Matrix' as defined in and established under the Retail Energy Code;
 - (b) the **"Data Permissions Matrix"** or **"DPM"** is the document of that name which sets out:
 - categories of data maintained by the CDSP in the Supply Point Register; and
 - (ii) for each such category, classes of person (not being Parties to the Code or parties to the IGT Code) to whom the CDSP is permitted to provide such data ("permitted data recipients");
 - (c) "Gas Enquiry Service" has the meaning given in the Retail Energy Code.
- 2.6.2 In this paragraph 2.6 references to data are to data maintained by the CDSP in the Supply Point Register pursuant to the provisions of the Code, the IGT Code, the Retail Energy Code or a Gas Transporter's Licence.
- 2.6.3 The Parties authorise the CDSP to provide data to permitted data recipients in accordance with the DPM, subject to the further provisions of this paragraph 2.6.
- 2.6.4 The IGT Code contains provisions equivalent to paragraph 2.6.3.
- 2.6.5 The DPM applies in respect of the provision of data by the CDSP to permitted data recipients including pursuant to any Third Party Services.
- 2.6.6 The terms on which the CDSP provides data to permitted data recipients shall be:
 - (a) in the case of data provided pursuant to the Retail Energy Code, the terms provided in or established pursuant to the Retail Energy Code;
 - (b) in the case of data provided pursuant to Third Party Services (including pursuant to the Gas Enquiry Service where provided by the CDSP), terms which comply with the Third Party and Additional Services Policy.
- 2.6.7 The CDSP shall not agree pursuant to Third Party Services to provide data to any person (other than a Core Customer) except:
 - (a) to the extent permitted in the DPM;
 - (b) to the extent required for the purposes of the CDSP complying with any contract between the CDSP and RECCo relating to provision of 'CDSP Further Services' (as defined in REC);

© 2005 all rights reserved

GTD 15

- (c) for the purposes of the Gas Enquiry Service (where provided by the CDSP) to the extent permitted in the DAM).
- 2.6.8 DPM may (and may only) be modified by a decision of the Contract Management Committee.

3 DATA SERVICES CONTRACT

3.1 Content of Data Services Contract

- 3.1.1 The DSC comprises:
 - (a) an agreement ("DSC Agreement") in accordance with paragraph 3.1.2;
 - (b) terms and conditions (**"DSC Terms and Conditions"**) in accordance with paragraph 3.1.3; and
 - (c) further documents ("CDSP Service Documents") in accordance with paragraph 3.1.4.
- 3.1.2 The DSC Agreement:
 - (a) is an agreement, to which the CDSP and all Parties are (or by accession become) party, by which the DSC Terms and Conditions are given effect and made binding between the CDSP and the Parties; and
 - (b) may be amended only by agreement of all of the parties to it.
- 3.1.3 The DSC Terms and Conditions:
 - (a) give effect to and make binding (as between the CDSP and each Party) each CDSP Service Document;
 - (b) oblige the CDSP to perform the Non-Service Functions assigned to it under the Code and the IGT Code;
 - incorporate and make binding (as between the CDSP and each Party) the CDSP-Related Provisions of the Code;
 - (d) provide for the CDSP to provide or otherwise procure CDSP Services effectively to help facilitate the efficient and integrated operation of the gas industry;
 - (e) contain terms and conditions of the DSC which are not contained in the CDSP Service Documents; and
 - (f) may only be amended by way of Code Modification in accordance with paragraph 2.5-; and
 - (f)(g) oblige the CDSP to ensure the costs incurred in the delivery of CDSP Services and the performance of Non-Service Functions are efficient and economic.

3.1.4 The CDSP Service Documents:

(a)	comprise the following documents:		
© 2005 all rights	reserved	GTD 16	Version 5.89 29 March 2023

(i) a description ("CDSP Service Description") of the CDSP Services, in

© 2005 all rights reserved

GTD 17

accordance with paragraph 3.2;

- a methodology ("Budget and Charging Methodology") for budgeting CDSP Costs and setting CDSP Charges, in accordance with paragraph 3.3;
- (iii) a policy ("Credit Policy") for Core Customer credit risk management and debt enforcement in relation to the DSC, in accordance with paragraph 3.4;
- (iv) the UK Link Manual in accordance with paragraph 5.2;
- (v) rules and procedures ("Change Management Procedures") for amending the CDSP Service Documents and for modifications of UK Link in accordance with paragraph 3.5;
- (vi) arrangements ("Contract Management Arrangements") for management and reporting in respect of the DSC in accordance with paragraph 3.6;
- (vii) a policy ("Third Party and Additional Services Policy") for Third Party Services and Additional Customer Services in accordance with paragraph 3.7; and
- (viii) the Transitional Arrangements Document in accordance with Part I paragraph 3 of the Transition Document, for so long as its provisions continue to be effective; and
- (b) may be amended in accordance with the provisions of the DSC Terms and Conditions and the Change Management Procedures.
- 3.1.5 Each document comprising part of the DSC shall be kept up to date and published by the Transporters (or the CDSP on their behalf) on the website of the Code Administrator.
- 3.1.6 The Parties shall keep the DSC under review to ensure (through amendment pursuant to the Change Management Procedure) that it continues to comply with the requirements of the Code.

3.2 CDSP Service Description

- 3.2.1 The CDSP Service Description shall set out a specification of each of the CDSP Services (other than Third Party Services and Additional Customer Services), based on the classification of Direct Services in paragraph 2.1.1 and Agency Services in paragraph 2.2.1, including a definition of the service and (in certain cases) a service level or service levels for performance of the service.
- 3.2.2 Under the CDSP Service Description:
 - (a) CDSP Services shall be divided into areas ("Service Areas") of related CDSP Services; and
 - (b) within each Service Area each individual CDSP Service is a "Service Line".

© 2005 all rights reserved

GTD 18

- 3.2.3 The CDSP Service Description shall (without limitation):
 - (a) oblige the CDSP to perform the functions assigned to it under the Code and the IGT Code as Code Services;
 - (b) in relation to a Code Service, specify the CDSP Service by reference to the relevant provision of the Code or the IGT Code, as the case may be (and, in the case of a Direct Service, by identifying the relevant function of the CDSP as specified in the Code or IGT Code);
 - (c) oblige the CDSP to perform the GRDA Functions.

3.3 Budget and Charging Methodology

- 3.3.1 The Budget and Charging Methodology shall provide for:
 - (a) the CDSP to prepare, consult on and publish a budget ("CDSP Budget") for each CDSP Year, identifying the activities expected to be performed by the CDSP in the CDSP Year and the forecast CDSP Costs in respect of those activities (and constituting the CDSP Annual Budget referred to in the CDSP Licence Condition);
 - (b) the basis for an appeal to the Authority in respect of the CDSP Budget as provided in the CDSP Licence Condition and the implementation of the Authority's decision;
 - a basis for the allocation of CDSP Costs to the activities of the CDSP, including the provision of CDSP Services;
 - (d) the basis on which the CDSP will determine CDSP Charges to Core Customers (and other persons) in respect of CDSP Services and other activities of the CDSP, with a view to recovering CDSP Costs and ensuring an adequate level of working capital, and consistent with the principles in paragraph 1.4.6;
 - (e) the CDSP to prepare and publish an annual charging statement ("Annual Charging Statement") for each CDSP Year setting out the CDSP Charges payable by Core Customers in respect of CDSP Services (other than Additional Customer Services); and
 - (f) the circumstances in which and basis on which the CDSP Budget may be amended within the CDSP Year, and the Annual Charging Statement and CDSP Charges amended as a result-<u>; and</u>
 - (f)(g) the information that must be included in the CDSP Budget.
- 3.3.2 The Budget and Charging Methodology shall provide for:
 - the CDSP to publish and keep under review the Budget and Charging Methodology and Annual Charging Statement;
 - (b) the Budget and Charging Methodology and Annual Charging Statement:
 - (i) to facilitate the objective of economic, efficient and transparent charging for the provision of CDSP Services; and

(ii) to achieve the Charging Methodology Objectives (as defined in the CDSP © 2005 all rights reserved GTD 19 Version 5.89 29 March 2023

Licence Condition);

(c) the further matters required under the CDSP Licence Condition.

3.3.3 The Budget and Charging Methodology shall provide for an allocation methodology ("Cost Allocation Methodology") for allocating CDSP Costs to activities of the CDSP for the purpose of deriving CDSP Charges which shall include:

- (a) the rules for allocating input costs used to establish the CDSP Budget to each CDSP activity:
- (b) specification of the sources of data required for the allocation of CDSP Costs; and
- (c) assumptions needed to allocate input costs used to establish the CDSP Budget.

3.4 Credit Policy

- 3.4.1 The Credit Policy shall provide for:
 - the basis on which the CDSP shall assess the creditworthiness of, and set an unsecured credit limit for, each Core Customer;
 - requirements for Core Customers to provide credit cover for indebtedness under the DSC in excess of the unsecured credit limit;
 - the determination of the outstanding indebtedness of Core Customers under the DSC;
 - (d) the steps to be taken where a Core Customer fails to comply with its credit obligations under the Credit Policy; and
 - (e) the steps to be taken where a Core Customer fails to pay an amount due under the DSC or is in other financial default of the DSC.

3.5 Change Management Procedures

- 3.5.1 The Change Management Procedures shall provide for:
 - (a) the basis on which a change or modification may be proposed:
 - (i) in respect of CDSP Services under the DSC;
 - (ii) in respect of UK Link;
 - (iii) in respect of the UK Link Manual; and
 - (iv) in respect of any other CDSP Service Document;
 - (b) the procedures to be followed in respect of any such proposed change or modification;
 - (c) the coordination of such procedures with the Modification Procedures, where the change to CDSP Services is required by a Code Modification; and

(d) the role of the Change Management Committee in connection with the © 2005 all rights reserved GTD 20 Version 5.89 29 March 2023 Formatted: List Paragraph, Outline numbered + Level: 4 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.6 cm + Indent at: 2.86 cm

foregoing.

3.5.2 The Change Management Procedures shall also provide for the procedure by which a CDSP Service Document will be modified pursuant to a Code Modification as provided in paragraph 2.5.

3.6 Contract Management Arrangements

3.6.1 The Contract Management Arrangements shall provide for:

© 2005 all rights reserved

GTD 21

- requirements as to the manner in which the CDSP meets its obligations to provide CDSP Services under the DSC;
- (b) the role of the Contract Management Committee in managing the DSC on behalf of Core Customers;
- (c) reporting and audit requirements in respect of the performance by the CDSP of its obligations under the DSC; and
- (d) procedures to be followed in respect of disputes arising under the DSC.

3.7 Third Party and Additional Services Policy

- 3.7.1 The Third Party and Additional Services Policy shall set out the basis on which the CDSP may provide and may agree to provide Additional Customer Services and Third Party Services.
- 3.7.2 The Third Party and Additional Services Policy shall include:
 - limits to the scope and amount of Third Party Services with a view to limiting risk and cost to Core Customers;
 - (b) principles governing the liability which the CDSP may accept in the provision of Third Party Services;
 - (c) the requirement to earn a margin (above an allocation of CDSP Costs) in the charges made for Third Party Services; and
 - (d) the basis on which CDSP may agree to provide Additional Services.

4 DSC COMMITTEES

4.1 Establishment and functions of DSC Committees

- 4.1.1 In connection with the DSC and the activities of the CDSP the following Network Code Sub-committees (each a "DSC Committee") are established:
 - (a) Change Management Committee; and
 - (b) Contract Management Committee.
- 4.1.2 The Change Management Committee shall perform the functions and have the powers and duties provided in this Section D and in the DSC in connection with changes to the DSC.
- 4.1.3 The Contract Management Committee shall perform the functions and have the powers and duties provided in this Section D and in the DSC in connection with the management of the DSC and review or approval of the activities of the CDSP.
- 4.1.4 A DSC Committee may establish a sub-committee for such purposes (within the scope of its functions, powers and duties) and comprising such members and on such terms as it decides; and references to a DSC Committee include any such sub-committee.

© 2005 all rights reserved

GTD 22

- 4.1.5 The DSC Committees are autonomous and (except pursuant to an appeal in accordance with paragraph 4.5 or otherwise as expressly provided in this Section D) the UNC Committee has no power to overrule a decision of a DSC Committee or reduce or qualify the scope of its functions, powers and duties.
- 4.1.6 A DSC Committee may decide to refer a matter (within the scope of its functions, powers and duties) to the other DSC Committee, but otherwise neither DSC Committee is subordinate to the other.
- 4.1.7 No decision of a DSC Committee may be made or (if made) shall be effective if the decision would cause the CDSP to be or act in breach of the DSC.
- 4.1.8 A DSC Committee shall have no function, power or duty in relation to the EPG Scheme or in relation to the CDSP acting as the EPG Scheme Administrator.

4.2 Constitution of DSC Committees

- 4.2.1 Each DSC Committee shall comprise representatives ("Committee Representatives") of each Customer Class as follows:
 - (a) 6 individuals appointed as representatives of Shipper Users ("Shipper User Representatives"); and
 - (b) 6 individuals appointed as representatives of Transporters and IGTs, of which:
 - (i) 2 shall be appointed by National Gas Transmission ("NTS Representative");
 - (ii) 2 shall be appointed by DN Operators ("DNO Representatives"); and
 - (iii) 2 shall be appointed by IGTs ("IGT Representatives").
- 4.2.2 Annex D-2 sets out the basis on which the Committee Representatives of each Customer Class are to be appointed and from time to time removed and/or replaced.
- 4.2.3 References to Committee Representatives of a Customer Group are to Committee Representatives appointed as representatives of the Customer Class or Customer Classes comprising that Customer Group.
- 4.2.4 Each meeting of a DSC Committee shall be chaired by a person (not being a Committee Representative or the Committee Secretary) nominated by the Code Administrator ("Committee Chairperson").
- 4.2.5 The Code Administrator shall nominate and may from time to time remove and replace) a person (not being a Committee Representative) as secretary (**"Committee Secretary"**) for each DSC Committee.
- 4.2.6 A person who is a director of the CDSP may not be a Committee Representative.
- 4.2.7 The same person may be a Committee Representative on both DSC Committees.
- 4.3 Voting arrangements

© 2005 all rights reserved

GTD 23

- 4.3.1 Each Committee Representative shall hold one (1) vote; provided that in the circumstances provided for in Annex D-2 a Committee Representative may hold more than one (1) vote.
- 4.3.2 A decision of a DSC Committee shall be made:
 - (a) where the DSC or this Section D expressly so provides, by the vote of the Committee Representatives of a particular Customer Group; and Committee Representatives of a Customer Class not comprised in that Customer Group shall not vote on the relevant matter; and
 - (b) except as provided in paragraph (a), by the vote of all Committee Representatives (a **"Full Vote"**).

and, in relation to any matter to be decided by a DSC Committee, the **"Voting Representatives"** are the Committee Representatives who (pursuant to this paragraph 4.3.1) are to vote on the matter.

- 4.3.3 If (after considering the views of the CDSP) there is a bona fide difference of opinion within a DSC Committee as to the question whether a matter falls within paragraph 4.3.2(a), that question shall be decided under paragraph 4.3.2(b), but that decision is subject to appeal under paragraph 4.5.
- 4.3.4 Decisions of a DSC Committee shall be made:
 - (a) where expressly so provided in the DSC or this Section D, by the unanimous vote of the Voting Representatives who are present at the relevant meeting and who vote in respect of the matter to be decided (a "Unanimous Vote"); and
 - (b) except as provided in paragraph (a) by a simple majority of the votes of the Voting Representatives who are present at the relevant meeting and who vote in respect of the matter to be decided,

and accordingly, where a Voting Representative is not present at the relevant meeting or abstains from voting in respect of a decision, such Voting Representative shall not be counted in determining whether a unanimous vote or (as the case may be) a simple majority of votes has been obtained.

- 4.3.5 Where on a vote conducted under paragraph 4.3.3(b), there is an equal number of votes in favour of and against the matter to be decided:
 - (a) no person shall exercise a casting vote; and
 - (b) for the purposes of the provisions of paragraph 4.5 in respect of appeal of decisions, the DSC Committee shall be treated as having made a decision against the relevant matter.
- 4.3.6 It is expected that decisions of DSC Committees will generally be consensual; and the person chairing a meeting of a DSC Committee may, in respect of any matter to be decided, invite any Voting Representative who objects to indicate their objection, on the basis that (in the absence of any such objection) the matter shall be treated as decided (by a Unanimous Vote).

© 2005 all rights reserved

GTD 24

4.4 Proceedings of DSC Committees

- 4.4.1 Unless in conflict with any provision of this paragraph 4, the provisions of Sections 4.5 and 5 (excluding Sections 5.1, 5.2, 5.5, 5.6, and 5.8) of the Modification Rules shall apply (mutatis mutandis, and disregarding references therein to other provisions of the Modification Rules) in relation to each DSC Committee.
- 4.4.2 Committee Representatives (of whom at least two (2) shall be Shipper Representatives representing at least two (2) Shipper Classes (excluding ANC allocated votes) and three (3) shall be DNO Representatives, NTS Representatives or IGT Representatives, excluding the Committee Chairperson) present at a meeting of a DSC Committee who can exercise six (6) votes shall be a quorum; provided a meeting convened at which the only decisions to be made fall within paragraph 4.3.2(a) in relation to a particular Customer Group shall be quorate notwithstanding the absence of such Committee Representatives provided all the Committee Representatives of the relevant Customer Group are present.
- 4.4.3 The Committee Chairperson may not make a determination (as provided in Section 5.10.1 of the Modification Rules) which would exclude a representative of a Party from attending or participating in the business of a meeting of a DSC Committee (but without prejudice to any other provision of this Section D or the DSC which limits the right of a Party to attend such a meeting).
- 4.4.4 The CDSP is entitled and required to attend (by one or more representatives) each meeting of a DSC Committee.
- 4.4.5 In relation to any matter to be considered by a DSC Committee, the "**next eligible**" meeting of the DSC Committee is the first meeting for which the matter can be included in the meeting's agenda in accordance with the requirements in paragraph 5 (including its provisions for meetings called on short notice, where applicable) of the Modification Rules.
- 4.4.6 A Committee Representative which is employed or otherwise engaged by a Core Customer shall not in general be precluded from participating in any meeting of a DSC Committee or voting on a matter in respect of which that Core Customer has an interest, but where:
 - (a) a decision to be taken by a DSC Committee concerns a dispute between the CDSP and a Core Customer, or a breach or a claim of breach by a Core Customer of the DSC; and
 - (b) the identity of the Core Customer is known to the DSC Committee,

then a Committee Representative, or alternate, employed or otherwise engaged by such Core Customer shall not vote on such decision.

4.5 Appeal

- 4.5.1 A Core Customer may appeal a decision of a DSC Committee subject to and in accordance with this paragraph 4.5.
- 4.5.2 The grounds on which a decision of a DSC Committee may be appealed are:

© 2005 all rights reserved

GTD 25

- (a) that the decision is not in accordance with a provision of this Section D or the DSC; or
- (b) that the decision is not consistent with the DSC Objectives.
- 4.5.3 A Core Customer appealing a decision of a DSC Committee must give notice to the Secretary of the appeal, within 10 Business Days after the publication of the decision, specifying the decision in question and the grounds (as provided in paragraph 4.5.2) on which the appeal is made.
- 4.5.4 Where notice of appeal of a decision is given in accordance with paragraph 4.5.3:
 - the Secretary shall send the notice to the relevant DSC Committee, the CDSP and each other Core Customer, and place the appeal on the agenda of the next meeting of the UNC Committee;
 - (b) any Core Customer may, within five Business Days after the notice of appeal was sent under paragraph (a), send to the UNC Committee representations in respect of the matter;
 - (c) the UNC Committee may request further information from the CDSP, the DSC Committee or any Core Customer in connection with the appeal;
 - (d) the UNC Committee shall decide the matter, by reference to the grounds of appeal in paragraph 4.5.2 (and consistent with the provisions of the DSC in respect of the decision in question), in one of the following ways:
 - (i) by upholding the decision of the DSC Committee;
 - by making a different decision in substitution for the DSC Committee's decision; and
 - (iii) by remitting the matter to the DSC Committee for a further decision with such guidance or instruction as the UNC Committee shall decide; and
 - (e) the UNC Committee shall at the same time (and regardless of which decision it makes within paragraph (d)) decide whether in its opinion the matter subject to appeal affects or may affect competition between Parties or any class of Parties, results or may result in a breach of a Relevant Licence Provision, or otherwise affects or may affect the interests of consumers, in each case to a material extent (the "further appeal criteria").
- 4.5.5 Subject to paragraphs 4.5.6, 4.5.7 and 4.5.8, the decision of the UNC Committee in respect of the appeal is final and binding.
- 4.5.6 Where, following the decision of the UNC Committee in respect of an appeal:
 - (a) a Core Customer considers that the grounds of appeal in paragraph 4.5.2 are met in respect of the UNC Committee's decision under paragraph 4.5.4(d); and
 - (b) the UNC Committee decided that any of the further appeal criteria in paragraph 4.5.5(e) were met,

the Core Customer may, within five Business Days after the publication of the UNC

© 2005 all rights reserved

GTD 26

Committee's decision, appeal to the Authority the UNC Committee's decision to the Secretary, by notice given to the Secretary setting out the basis on which it considers the grounds of appeal in paragraph 4.5.2 are met.

- 4.5.7 Where a Core Customer gives notice of appeal of the UNC Committee's decision, the matter shall be referred to the Authority as follows:
 - the Secretary shall send to the Authority a statement of the decision subject to appeal together with relevant papers which were considered by the UNC Committee in reaching its decision;
 - (b) any Core Customer may, within five Business Days after the statement in paragraph (a) is published, send to the Authority representations in respect of the matter; and
 - (c) the Authority is requested to determine the matter, by reference to the grounds of appeal in paragraph 4.5.2, in one of the following ways:
 - (i) by upholding the decision of the UNC Committee;
 - by making a different decision in substitution for the UNC Committee's decision; or
 - (iii) by remitting the matter to the UNC Committee for a further decision with such guidance or instruction as the Authority shall decide.
- 4.5.8 The decision of the Authority in respect of an appeal is final and binding.
- 4.5.9 Until the UNC Committee or (if its decision is appealed to the Authority) the Authority makes a determination in respect of an appeal, the decision of the DSC Committee stands; but where a decision is appealed the DSC Committee shall make a separate decision (under paragraph 4.3.1(b)) as to whether the CDSP is to act on the decision pending the decision of the UNC Committee and (where applicable) the decision of the Authority.
- 4.5.10 The provisions of this paragraph 4.5 for appeal of a decision of a DSC Committee are without prejudice to the ability of a Core Customer to assert that a purported decision of a DSC Committee is outside its scope of authority pursuant to this Section D and the DSC and so is without effect.

5 UK LINK

5.1 Introduction

- 5.1.1 This paragraph 5 sets out:
 - (a) requirements of the Code in respect of UK Link and the UK Link Manual; and
 - (b) the basis on which Code Communications are to be given as UK Link Communications.
- 5.1.2 For the purposes of the Code:
 - (a) "UK Link" means the information technology and communications (ITC)

© 2005 all rights reserved

GTD 27

systems (in aggregate) used by the CDSP to provide CDSP Services (including an information exchange system allowing the electronic transfer of information between the CDSP and UK Link Users and access to other parts of such ITC systems), as described in the UK Link Manual, and as from time to time modified in accordance with the Change Management Procedures;

- (b) "UK Link Gemini" means those parts of UK Link which support the implementation of the provisions of TPD Section B in relation to NTS Entry Capacity and NTS Exit Capacity and of TPD Sections E and F in respect of the determination of Energy Balancing Charges, as further specified in the UK Link Manual;
- (c) a "UK Link User" is:
 - (i) the CDSP;
 - any Party in its capacity as a user of UK Link, provided that a Discontinuing User shall cease to be a UK Link User upon the User Discontinuance Date;
 - (iii) the CSS Provider for the purposes provided for in the Retail Energy Code;
 - (iv) any other person permitted to have access to and use of UK Link in accordance with the Code, the UK Link Manual, and/or the Third Party and Additional Services Policy;
- (d) a "UK Link Communication" is a message transmitted by a UK Link User by means of UK Link in accordance with the requirements of the UK Link Manual; and for these purposes a "message" is a discrete transmission electronically sent or electronically posted by a UK Link User, and includes a message accessible in UK Link by API (application programmable interface);
- (e) a "Code Communication" is:
 - (i) a TPD Communication;
 - an IGTAD Communication (excluding an IGTAD Communication given pursuant to IGTAD Section B); and
 - (iii) for the purposes of the provisions of this paragraph 5 relating to the UK Link Manual, an IGT Code Communication which is stated in the IGT Code to be a Code Communication; and
- (f) a "UK Link User Agreement" is an agreement between the CDSP and another UK Link User other than a Core Customer which provides for such person to have access to and use of UK Link or any part of it.
- 5.1.3 The CDSP will secure the establishment and operation of UK Link, and will afford to UK Link Users access to and use of UK Link, subject to paragraph 5.1.4.

5.1.4 In relation to UK Link Gemini:

(a) the CDSP maintains UK Link Gemini on behalf of National Gas Transmission;

© 2005 all rights reserved

GTD 28

- (b) National Gas Transmission will afford to UK Link Users access to and use of UK Link Gemini; and
- (c) accordingly the CDSP makes UK Link Gemini available and performs other functions in relation to UK Link Gemini (in accordance with the relevant Service Lines in the CDSP Service Description) as Agency Services for National Gas Transmission.
- 5.1.5 Parties and the CDSP will communicate with each other by means of UK Link, subject to and in accordance with this paragraph 5.
- 5.1.6 Parties and the CDSP are required to comply with the relevant requirements of this paragraph 5 and the UK Link Manual in respect of access to and use of UK Link.
- 5.1.7 Annex D-1 sets out provisions relating to facilities within UK Link available for trading between Users.
- 5.1.8 The IGT Code contains provisions equivalent to provisions of this paragraph 5 which relate to sending and receiving UK Link Communications.

5.2 UK Link Manual

- 5.2.1 The "UK Link Manual" is the document of that title containing the following matters:
 - (a) a description of UK Link (or each part of UK Link);
 - (b) terms and conditions on which UK Link Users have access to and use of UK Link;
 - (c) details of how UK Link will be made available to UK Link Users, and when it may not be available;
 - (d) details of the means by which Code Communications are to be made; and (in relation to Code Communications to be made as UK Link Communications) details of how such UK Link Communications are made and recorded, including the matters set out in paragraph 5.6 and 5.8;
 - (e) the Code Contingency Guidelines Document as referred to in paragraph 5.9; and
 - (f) such further matters as are provided for in this paragraph 5 or elsewhere in the Code, or as the CDSP considers and the Change Management Committee or (in a case within paragraph 5.2.3) the UNC Committee accepts are appropriate.
 - (g) a document prepared and managed by the CDSP which sets out the data items available to the Parties (the **"Data Permissions Matrix"**).
- 5.2.2 The UK Link Manual is a CDSP Service Document for the purposes of this Section D and the DSC, and is a UNC Related Document for the purposes of the Code.
- 5.2.3 It shall be an obligation of a UK Link User under the Code to comply with a provision of the UK Link Manual where such provision is expressly identified in the UK Link Manual as one which is made binding on such UK Link User by this paragraph 5.2.3, and not otherwise; but it is acknowledged that as respects all provisions of the UK Link Manual (whether or not made binding by this paragraph 5.2.3) a UK Link User may be

© 2005 all rights reserved

GTD 29

unable to make a UK Link Communication (and so may be unable to exercise an entitlement to make a Code Communication required to be made as such) where the UK Link User does not comply with such provisions.

5.2.4 The UK Link Manual does not form a part of the Code (without prejudice to paragraph 5.2.3); and, in case of any conflict between the Code and the UK Link Manual, the Code shall prevail.

5.3 Code Communications

- 5.3.1 A UK Link Communication given in accordance with this paragraph 5.3 and the UK Link Manual shall be treated as an effective and valid Code Communication, and each Party confirms that it intends and agrees that UK Link Communications shall have legal effect for the purposes of the Code.
- 5.3.2 The UK Link Manual specifies in respect of each Code Communication therein listed whether it is to be given as a UK Link Communication, by Active Notification System, by Conventional Notice, by e-mail, by facsimile or by telephone, and in some cases alternative such means by which it may be given; and (subject to paragraph 5.2.4 and GT Section B5.1.2) a Code Communication may only be given by the means so specified or (where alternative such means are specified) by one of such alternative means.
- 5.3.3 Where the Code or the UK Link Manual specifies the form and/or format of UK Link Communication by which a particular Code Communication is required to be given, that Code Communication may be given only in that form and/or that format.
- 5.3.4 In the event of a Code Contingency, Code Communications shall be (and are permitted to be) given on the basis set out in paragraph 5.9.
- 5.3.5 Where a Code Communication which is required to be given as a UK Link Communication (and is not permitted to be given by another means except pursuant to paragraph 5.3.4) is not given in accordance with the requirements of this paragraph 5.3 and the UK Link Manual, and as a result is not properly received by the intended recipient, it shall be deemed not to have been given and shall be of no effect.
- 5.3.6 The failure of a UK Link User to comply with a requirement that a Code Communication be given as a UK Link Communication, or as to the form or format in which such UK Link Communication is to be given, shall not of itself be a breach of the Code (but without prejudice to paragraph 5.3.5 or to any breach which may result from the failure to give the Code Communication).

5.4 Planned UK Link downtime

- 5.4.1 Pursuant to the UK Link Manual:
 - (a) on each Day and/or particular Days UK Link, or (where so specified in the UK Link Manual) particular parts of UK Link, will not be operational at certain times and for certain periods ("planned UK Link downtime") specified in or determined in accordance with the UK Link Manual; and
 - (b) during planned UK Link downtime,

© 2005 all rights reserved

GTD 30

UK Link Users will not be able to have access to or use UK Link or the relevant part thereof, and accordingly will not be able to make UK Link Communications, except to the extent, if any, provided for in the DSC or the UK Link Manual.

- 5.4.2 During planned UK Link downtime, notwithstanding any other provision of the Code, Parties will not be entitled to make any Code Communication which is required to be made as a UK Link Communication, except to the extent (if any) provided for in the UK Link Manual or (where part only of UK Link is subject to such downtime) where such part is not required to enable such communication to be made, and their respective rights under the Code shall be construed accordingly.
- 5.4.3 Except as provided in paragraph 6.5(b) of the UK Link Terms and Conditions (as defined in the DSC), the unavailability of UK Link during planned UK Link downtime will not be a Code Contingency for the purposes of paragraph 5.9; but any such unavailability which extends or the CDSP reasonably anticipates will extend beyond the period of planned UK Link downtime will (subject to and in accordance with the Contingency Procedures) be a Code Contingency.

5.5 Access to and use of UK Link

- 5.5.1 For the purposes of the Code:
 - the UK Link Manual sets out requirements ("UK Link Access Requirements") which UK Link Users must comply with in order to have access to and use UK Link;
 - (b) (without prejudice to the DSC) it is a requirement for the purposes of the Code that each Party complies with the UK Link Access Requirements; and
 - (c) where a Party does not comply with the UK Link Access Requirements, the Party may be unable to give or receive Code Communications, and (unless otherwise provided) such inability shall not relieve the Party from any obligation or liability under the Code.
- 5.5.2 The UK Link Access Requirements include requirements for the provision by the CDSP of certain facilities and services to enable a UK Link User to have access to UK Link:
 - (a) which are provided as Direct Services as specified in the CDSP Service Description; and
 - (b) for which the CDSP Charges are to be determined on a uniform basis for Core Customers, Trader Users and UK Link Users who are not Parties.
- 5.5.3 The security arrangements set out in the UK Link Manual include the basis on which a UK Link User may nominate representatives as authorised to access and use UK Link on behalf of and using the identification of that UK Link User.
- 5.5.4 Any other UK Link User shall be entitled to assume that any person nominated as such a representative by a UK Link User or otherwise using the identification of the UK Link User is fully authorised to access and use UK Link (and in particular to initiate, authorise and transmit, and to receive or access for the purposes of receiving, UK Link Communications); and any UK Link Communication transmitted by such a person shall be treated as given by the UK Link User.

© 2005 all rights reserved

GTD 31

5.5.5 In any case (including a Code Contingency) where the CDSP suspends or inhibits access (by a UK Link User or UK Link Users) to UK Link in accordance with the DSC, such UK Link User(s) may be prohibited from using UK Link; and it is acknowledged that, in such circumstances, such UK Link User(s) may be unable to make a UK Link Communication (and so may be unable to exercise an entitlement to make a Code Communication required to be made as such).

5.6 UK Link Communications

- 5.6.1 The UK Link Manual specifies:
 - (a) different forms of UK Link Communication; and
 - (b) for each Code Communication required to be given as a UK Link Communication, the required form of UK Link Communication.
- 5.6.2 For each form of UK Link Communication, the UK Link Manual specifies (in respect of messages comprised in that form of UK Link Communication):
 - the class or classes of UK Link User by which and to which each message is sent;
 - (b) in certain cases, times at which and/or periods within which such messages may be sent;
 - (c) the basis (if any) on which the message is validated as complying with applicable requirements of the UK Link Manual and may be rejected by UK Link where invalid;
 - (d) the basis on which and the facility or system (within UK Link) in which the sending or receipt of the message will be recorded;
 - (e) the basis on which the time and date of sending or receipt of the message will be recorded;
 - (f) whether the message is treated as received by the recipient when recorded as sent or as received;
 - (g) whether a recipient of the message must access the message within UK Link after the message has been sent and treated as received;
 - (h) in a case within paragraph (g), the basis on which the recipient(s) of the message can access the message, and the period within which they may do so (and the basis on which the message may or will be removed after such period and following which the message is or may be no longer accessible);
 - whether a message acknowledging receipt of the UK Link Communication is automatically generated and sent; and
 - (j) in a case within paragraph (i), procedures to be followed by the sender and intended recipient where no such acknowledgment is received.
- 5.6.3 The provisions of the UK Link Manual referred to in paragraph 5.6.2 are effective for the purposes of the Code; and accordingly where any Code Communication is sent as a

© 2005 all rights reserved

GTD 32

UK Link Communication:

- the Code Communication may be given only in accordance with those provisions;
- (b) the Code Communication shall be deemed given and received in accordance with those provisions; and
- (c) any question as to whether and when the Code Communication was effectively given or received shall be determined in accordance with those provisions.
- 5.6.4 In any case referred to in paragraph 5.6.2(g):
 - (a) each UK Link User which is a recipient of the UK Link Communication shall be responsible for accessing the relevant part of UK Link (to access such messages) at such intervals as shall be appropriate and prudent; and
 - (b) the relevant Code Communication shall be deemed to have been received by a UK Link User to whom it is transmitted at the time determined in accordance with the UK Link Manual, irrespective of whether or when accessed (as referred to in Section 5.6.2(h)) by that UK Link User.
- 5.6.5 In any case referred to in paragraph 5.6.2(j):
 - (a) the relevant UK Link Users shall comply with the procedures in the UK Link Manual; and
 - (b) the question whether the message is treated as received by the intended recipient shall be determined, by reference to those procedures, as set out in the UK Link Manual.

5.7 Active Notification

- 5.7.1 An **"Active Notification Communication"** is a message transmitted by National Gas Transmission by means of the Active Notification System, as described in the UK Link Manual.
- 5.7.2 The **"Active Notification System"** is the system (as described in the UK Link Manual) provided and operated by National Gas Transmission for transmitting messages to a warning device (**"Active Notification Device"**).
- 5.7.3 The Active Notification System does not form part of UK Link and the CDSP has no functions in relation to the Active Notification System.
- 5.7.4 An Active Notification Communication shall be deemed to have been received by the UK Link User to whom it is transmitted at the time it is logged 'as sent' by the Active Notification System.
- 5.7.5 Each UK Link User (of a class to which Active Notification Communications may be sent) shall ensure that at all times a representative of the UK Link User has an Active Notification Device in their continuous possession and control.
- 5.7.6 An Active Notification Communication may (where so provided in the UK Link Manual) be notice to the effect that a Code Communication has been given by National

© 2005 all rights reserved

GTD 33

Gas Transmission by some other permitted means (in which case the Active Notification Communication will not contain the Code Communication itself).

5.8 Audit Trail

- 5.8.1 The CDSP will retain a complete and chronological record of all UK Link Communications transmitted received by means of UK Link, for the minimum period applicable in respect of the relevant communication, as specified in the UK Link Manual, following transmission or receipt.
- 5.8.2 The CDSP (in accordance with the DSC) and each other UK Link User shall ensure that electronic or computer records containing UK Link Communications are readily accessible and are capable of being reproduced in a human readable form and of being printed, if required.
- 5.8.3 Records made by the facilities or systems (as referred to in paragraph 5.6.2(d)) within UK Link of the transmission or receipt of UK Link Communications, and (as respects UK Link Communications the transmission or receipt of which is not so recorded) other records retained by the CDSP and/or UK Link Users in accordance with this paragraph 5.8, shall be prima facie evidence of the transmission or receipt of such UK Link Communications.
- 5.8.4 In the event of a dispute between any UK Link Users as to any Code Communication the sending of which was (as a UK Link Communication) recorded by UK Link, the CDSP will, as soon as reasonably practicable after a request to do so, provide to the UK Link User a copy of what is recorded (in respect of such communication) in such facility.

5.9 Contingency Arrangements

- 5.9.1 For the purposes of the Code:
 - the "Contingency Procedures" are procedures contained in the Code Contingency Guidelines Document forming part of the UK Link Manual for the UK Link Users to communicate with each other in the event of a Code Contingency;
 - (b) a "Code Contingency" is an event or circumstance affecting UK Link, of a kind specified in the Contingency Procedures, which affects the ability of UK Link Users (or where so specified in the Contingency Procedures, of a particular UK Link User or particular UK Link Users) to give or receive UK Link Communications, or to generate information to be contained in a Code Communication; and
 - (c) the "Code Contingency Guidelines Document" is the document so titled forming part of the UK Link Manual and containing consolidated guidelines in respect of Code Contingencies.
- 5.9.2 The Parties and the CDSP adopt and (in the event of a Code Contingency) shall implement the relevant Contingency Procedures.
- 5.9.3 In the event of a Code Contingency, where so provided in the Contingency Procedures, a Code Communication which would normally be required to be given as a UK Link

© 2005 all rights reserved

GTD 34

Communication may (notwithstanding any other provision of the Code or the UK Link Manual) be given by any means provided for in the Contingency Procedures.

- 5.9.4 Except as provided in the Code or the Contingency Procedures, the provisions of the Code will apply in and will not be affected by a Code Contingency; and in particular any requirements under the Code or the UK Link Manual as to the timing and content of any Code Communication, and the giving of communications by means other than UK Link, will continue to apply.
- 5.9.5 Under the Contingency Procedures communication resources of UK Link Users will be used for the purposes of facilitating the continued giving (in accordance with the Contingency Procedures) of certain kinds of Code Communications in a Code Contingency, and so may not be available for other kinds of Code Communications; and accordingly, in any case where the Contingency Procedures do not make provision for the giving of a particular kind of Code Communication (which is required to be given as a UK Link Communication), it may not be possible for Code Communications of that kind to be given.
- 5.9.6 It is agreed that where certain Code Contingencies occur or continue for particular periods or at particular times:
 - (a) the application of certain provisions of the Code (in particular, the requirement for Users to pay certain charges, or the basis on which such charges are determined), will be modified; and/or
 - (b) the timetable provided for in the Code for the giving of certain Code Communications may be extended,

as provided in the relevant Section of the Code.

- 5.9.7 A Code Contingency of the kind referred to in paragraph 5.9.6 is a "Class A Contingency".
- 5.9.8 Code Contingencies which are Class A Contingencies are specified, for the purposes of the relevant provisions of the Code, in the Contingency Procedures and referenced where appropriate in the relevant Sections of the Code.

6 MARKET PARTICIPANT REGISTRATION

6.1 Introduction

- 6.1.1 This paragraph 6 sets out the requirements of the Code with respect to the registration of and assignment of identity codes to, Market Participants by the CDSP ("Market Participant Identity Registration Services").
- 6.1.2 Market Participant Identity Registration Services are Direct Functions of the CDSP.
- 6.1.3 For the purposes of the Code, a Market Participant is any one of the following;
 - (a) a supplier;
 - (b) a person who holds a Shipper's Licence (whether or not a Shipper User);
 - (c) a Meter Asset Manager;

© 2005 all rights reserved

GTD 35

- (d) a person who makes a Supply Meter Installation available for use by a supplier and consumer (a "Meter Asset Provider"); and
- (e) the holder of a Gas Transporter's licence.

and for such purposes, Section A1.5 of the Independent Gas Transporter Arrangements Document shall apply.

- 6.1.4 The UK Link Manual shall incorporate a document (the **"Market Participant Identity Verification Approach Document"**) setting out the processes to be followed and verification measures to be applied by the CDSP in compiling, updating and amending the Market Participants List.
- 6.1.5 For the purposes of the Code, a **"Market Participant ID**" is a unique code for a Market Participant identifying it, the type (or types) of Market Participant that it is and such other information in relation to it as is provided for, and in the form provided for, by the Market Participant Identity Verification Approach Document.

6.2 The Market Participant Identity Registration Services

- 6.2.1 The CDSP shall compile, maintain and publish the **"Market Participant Identities** List" being:
 - (a) a record of all Market Participants which have applied to the CDSP for a Market Participant ID to be assigned to them, and such other Market Participants as may be provided for by the Market Participant Identity Verification Approach Document; and
 - (b) the Market Participant IDs assigned to each such Market Participant from time to time.
- 6.2.2 The CDSP shall publish such information as may be reasonably required by Market Participants which are not Parties for the purposes of enabling them to acquire a Market Participant ID and secure its amendment as appropriate from time to time.
- 6.2.3 The CDSP shall provide to the Smart Energy Code Panel such information, support and assistance as it may reasonably request for the purposes of exercising its function of making a determination under either Section L3.29 or Section L3.30 of the Smart Energy Code, and shall provide to the Authority such information, support and assistance as it may reasonably request for the purposes of exercising its function of deciding any appeal brought under Section L3.33 of the Smart Energy Code.

© 2005 all rights reserved

GTD 36

ANNEX D-1 - TRADING BETWEEN USERS

1 General

1.1 Users may use the UK Link Network (in addition to use for communicating with the Transporter) for the purposes of making User Trade Communications, in accordance with and subject to the relevant provisions of the UK Link Manual.

1.2 A "User Trade Communication" is:

- (a) an invitation by a User to other Users to make such an offer as is referred to in paragraph (b);
- (b) an offer by a User to another User to make:
 - (i) a System Capacity Transfer, as Transferee User; or
 - (ii) a Storage Transfer, as transferee; or
- (c) an acceptance by a User of such an offer (made by another User) as is referred to in paragraph (b) (such Users, the **"Trading Users"**).
- 1.3 A User Trade Communication shall be a UK Link Communication but (without prejudice to paragraph 4.1) is not a Code Communication.
- 1.4 Without prejudice to any other provision of the Code, nothing in this paragraph 1 prevents any User agreeing with another User a Capacity Transfer, Trade Nomination or Storage Transfer other than pursuant to a User Trade Communication.

2 Basis of User Trade Communications

- 2.1 A User Trade Communication shall be made and recorded as provided in the UK Link Manual.
- 2.2 The UK Link Manual sets out:
 - (a) the basis on which a User Trade Communication may be made, and restrictions applying thereto; and
 - (b) details of the information concerning the User making the User Trade Communication, or comprised in that communication, which will be available to or accessible by other Users by or on the UK Link Network.

3 Effect of User Trade Communications: Users

- 3.1 Each User agrees with each other User that:
 - (a) a User Trade Communication shall be treated as a valid and effective communication as between Users;
 - (b) a User Trade Communication under paragraph 1.2(b) shall be an offer capable

© 2005 all rights reserved

GTD 37

of acceptance by the User to whom it is addressed; and

- (c) a User Trade Communication under paragraph 1.2(c) shall be an acceptance of an offer giving rise to a contract between the relevant Users.
- 3.2 The terms of the contract referred to in paragraph 3.1(c) shall be those terms contained in the relevant User Trade Communication(s) and such other terms as shall have been agreed by the Trading Users or in default of such agreement the terms in the Appendix to this Annex D-1.
- 3.3 GT Section B2.4.1 shall not apply in respect of paragraphs 3.1 and 3.2.

4 Effect of User Trade Communication: Transporter

- 4.1 Upon a User making a User Trade Communication under paragraph 1.2(c), the Trade Notices will (as described in the UK Link Manual) automatically be given, and shall be treated as having been given for all purposes of the Code, by that User and the User who made the corresponding User Trade Communication under paragraph 1.2(b).
- 4.2 For the purposes of paragraph 4.1, the **"Trade Notices"** are:
 - (a) in the case of a System Capacity Transfer, the notices required under TPD Section B5.2.1; and
 - (b) in case of a Storage Gas Transfer, the notices required by the relevant Storage Terms.

GTD 38

APPENDIX TO ANNEX D-1

1 Application of this Appendix

- 1.1 This Annex does not form part of the Uniform Network Code.
- 1.2 This Annex sets out certain terms which, subject to paragraph 1.3, are incorporated into every contract arising as a result of a User Trade Communication made under Annex D-1 paragraph 1.2(c) (a **"User Trade Contract"**).
- 1.3 In accordance with Annex D-1 paragraph 3.2, a term set out in this Annex is not incorporated into a User Trade Contract:
 - (a) if the Trading Users have expressly agreed another term; or
 - (b) to the extent that such term is inconsistent with any term expressly agreed by the Trading Users.

2 Interpretation

- 2.1 Terms used in this Annex have the same meanings as in the Code.
- 2.2 For the purposes of this Annex:
 - (a) **"applicable daily rate"** means:
 - (i) in respect of System Capacity, the Applicable Daily Rate (in pence per kWh/Day) of the relevant Capacity Charge;
 - (ii) in respect of Storage Space, the Applicable Storage Space Charge Rate (in pence per kWh of Storage Space for a Storage Year) divided by 365;
 - (iii) in respect of Storage Deliverability, the Applicable Storage Deliverability Charge Rate (in pence per kWh/Day of Storage Deliverability for a Storage Year) divided by 365; and
 - (iv) in respect of Storage Injectability, the Applicable Storage Injectability Charge Rate (in pence per kWh/Day of Storage Injectability for a Storage Year) divided by 365;
 - (b) **"Capacity Transfer"** means a System Capacity Transfer or a Storage Capacity Transfer;
 - (c) "Capacity Transfer Day" means a Day in the Relevant Transfer Period in respect of which the Trade Transferee is treated as holding the Transferred Capacity in accordance with the Code;
 - (d) "Relevant Transfer Period" means:

© 2005 all rights reserved

GTD 39

- (i) in respect of a System Capacity Transfer the Transfer Period; and
- (ii) in respect of a Storage Capacity Transfer the Storage Transfer Period;

(e) **"Trade Transferor"** means:

- (i) in respect of a System Capacity Transfer the Transferor User; and
- (ii) in respect of a Storage Transfer the Transferor Storage User;
- (f) **"Trade Transferee"** means:
 - (i) in respect of a System Capacity Transfer the Transferee User; and
 - (ii) in respect of a Storage Transfer the Transferee Storage User; and
- (g) **"Transferred Capacity"** means Transferred System Capacity or Transferred Storage Capacity.

3 Price

3.1 Capacity Transfer

3.1.1 In respect of a Capacity Transfer the Trade Transferee shall, subject to paragraph 3.1.2, pay to the Trade Transferor in respect of each month some or all of the Days of which are Capacity Transfer Days a sum determined in accordance with the following formula:

A = B * C * D

where

- A = the sum payable by the Trade Transferee in respect of the month;
- B = the applicable daily rate in respect of the Capacity Transfer;
- C = the amount of the Transferred Capacity; and
- D = the number of Capacity Transfer Days in that month.
- 3.1.2 If in respect of any Capacity Transfer Day in a month the Trade Transferor ceases to be liable to pay to National Gas Transmission or National Gas Transmission LNG Storage the relevant Capacity Charges or Storage Capacity Charges in respect of an amount of Transferred Capacity in accordance with the Code, the sum which would but for this paragraph 3.1.2 be payable by the Trade Transferee pursuant to paragraph 3.1.1 in respect of that month shall be reduced by a sum equal to the aggregate of the sums for each such Capacity Transfer Day, each such sum being calculated in accordance with the following formula:

$$E = F * G$$

where:

E = is the sum by which the Trade Transferee's liability is reduced in respect of such Day;

© 2005 all rights reserved

GTD 40

- F = is the applicable daily rate; and
- G = is the amount of Transferred Capacity in respect of which the Trade Transferror ceased to be liable to pay Capacity Charges or Storage Capacity Charges in respect of such Day.

3.2 Storage Gas Transfer

3.2.1 In respect of a Storage Gas Transfer the Trade Transferee shall pay to the Trade Transferor a sum determined in accordance with the following formula:

$$H = I * J$$

where:

- H = is the sum payable by the Trade Transferee in respect of the Storage Gas Transfer;
- I = the Operating Margins WACOG (in pence per kWh) for the Transfer Storage Facility on the date the Operating Margins WACOG was last calculated before the transfer date; and
- J = is the Transferred Gas-in-Storage (in kWh).

4 VAT

The sums payable pursuant to a User Trade Contract are exclusive of Value Added Tax.

5 Payment

- 5.1 Sums payable by the Trade Transferee to the Trade Transferor under a User Trade Contract shall be paid on or before whichever is the later of:
 - (a) the 5th Day after the Day on which the Trade Transferee receives an invoice in respect of such sum from the Trade Transferor; and
 - (b) the 15th Day of the month following the relevant month (in accordance with paragraph 5.2).
- 5.2 For the purposes of paragraph 5.1 the relevant month is:
 - (a) for a Capacity Transfer, the month to which the invoice relates; and
 - (b) for a Storage Gas Transfer, the month in which the User Trade Communication under Annex D-1 paragraph 1.2(c) was made.

6 Interest

6.1 Where any sum payable pursuant to a User Trade Contract is not paid on or before the due date for payment in accordance with paragraph 5.1 (the "Due Date"), the paying User shall pay interest, after as well as before judgment, at the rate for the time being of Barclays Bank PLC plus 3 percentage points per annum on the unpaid amount from the Due Date until the day on which payment is made.

© 2005 all rights reserved

GTD 41

6.2 Interest payable pursuant to paragraph 6.1 shall accrue on a daily basis and on the basis of a 365-day year.

7 Governing law and jurisdiction

- 7.1 Each User Trade Contract shall be governed by, and construed in all respects in accordance with, English law.
- 7.2 The parties to a User Trade Contract irrevocably agree that the courts of England are to have exclusive jurisdiction to settle any disputes which may arise out of or in connection with that User Trade Contract.

© 2005 all rights reserved

ANNEX D-2 - COMMITTEE REPRESENTATIVES

1 Defined terms

- 1.1 For the purposes of this Annex D-2:
 - (a) **"Appointment Year"** means the period from 1 October in any year until and including 30 September in the following year;
 - (b) "Nominating Shipper User" means:
 - (i) a Shipper User; or
 - where there is more than one relevant Shipper User in a Shipper User Group, a Shipper User which has been authorised by all Shipper Users in that group to act on their behalf under this Annex D-2;
 - (c) "Relevant Date" means 30 June in Appointment Year Y-1;
 - (d) **"Shipper User Group"** means a Shipper User, and where another Shipper User is a fifty percent (50%) Affiliate of the Shipper User, such Shipper Users taken together, and in relation to a Shipper User Group:
 - a "relevant" Shipper User is a Shipper User belonging to the Shipper User Group;
 - the "relevant" Class is the Class to which the Shipper User Group (and its Nominating Shipper User) belongs.

2 Shipper User Representatives

2.1 Classes of Shipper User Groups

- 2.1.1 A Shipper User Group is in Class A if at the Relevant Date the number of Supply Points in respect of which a relevant Shipper User is the Registered User exceeds, in aggregate, one million (1,000,000).
- 2.1.2 A Shipper User Group is in Class B if at the Relevant Date:
 - (a) it is not a Class A Shipper User Group; and
 - (b) in respect of Supply Points in relation to which a relevant Shipper User is the Registered User, not less than fifty per cent (50%) of such Supply Meter Points comprised in such Supply Points are assigned a Market Sector Code indicating the Supply Point Premises are Non-domestic Premises.
- 2.1.3 A Shipper User Group is in Class C if it is not in Class A or in Class B.

2.2 Appointment

© 2005 all rights reserved

GTD 43

- 2.2.1 For each Appointment Year (Y) the Shipper User Representatives shall be appointed in accordance with this paragraph 2.2.
- 2.2.2 Subject to the further provisions of this paragraph 2.2, the Shipper User Representatives for each DSC Committee shall be appointed as follows:
 - (a) two (2) representatives shall be appointed by Shipper User Groups in Class A;
 - (b) two (2) representatives shall be appointed by Shipper User Groups in Class B;
 - (c) two (2) representatives shall be appointed by Shipper User Groups in Class C.
- 2.2.3 For each Appointment Year (Y), and for each DSC Committee a Shipper User Group may nominate one (1) individual as a candidate ("**Candidate**") for election as a representative of the relevant Class, in accordance with paragraph 2.2.4.
- 2.2.4 For each Shipper User Group and for each DSC Committee, the Nominating Shipper User shall submit to the Code Administrator:
 - (a) the name of the Candidate (if any) nominated by it (together with evidence of the Candidate's willingness to act as a representative) by no later than 21 July in Appointment Year Y-1; and
 - (b) at the same time, a declaration that either:
 - (i) the Nominating Shipper User is the only Shipper User in the relevant Shipper User Group; or
 - the Nominating Shipper User is authorised by all Shipper Users in the relevant Shipper User Group to act on their behalf for the purposes of this Annex D-2, and such declaration shall identify the names of all Shipper Users in the Shipper User Group;

and a Shipper User Group for which a submission is not made under this paragraph 2.2.4 shall not participate further in the appointment process for the relevant Appointment Year.

- 2.2.5 Following receipt of nominations under paragraph 2.2.4:
 - (a) the Code Administrator shall send to the CDSP the details of Shipper User Groups submitting a nomination pursuant to paragraph 2.2.4;
 - (b) the CDSP shall determine and notify to the Code Administrator the Class in which each Shipper User Group falls (in accordance with paragraph 2.1).
- 2.2.6 By no later than 31 July in Appointment Year Y-1 the Code Administrator shall:
 - (a) notify each Nominating Shipper User of:
 - the Class in which the Nominating Shipper User's Shipper User Group falls;
 - (ii) the names of all Candidates nominated by Shipper User Groups in the relevant Class; and

© 2005 all rights reserved

GTD 44

- (b) where the number of Candidates for a Class is greater than two (2):
 - (i) invite each of the Nominating Shipper Users in the relevant Class to vote for three (3) of the Candidates;
 - (ii) notify the Nominating Shipper Users of the date by which votes are to be submitted to the Code Administrator for the purposes of 2.2.8(d).
- 2.2.7 Where paragraph 2.2.6(b) applies in respect of a Class, each Nominating Shipper User in such Class shall vote, by written notice to the Code Administrator (in such form as the Code Administrator may require) submitted by no later than the date referred to in paragraph 2.2.6(b)(ii), for three (3) of the Candidates for the relevant Class.
- 2.2.8 For each DSC Committee, and for each Class of Shipper User Group, the Shipper User Representatives shall be appointed as follows (where ANC is the aggregate number of Candidates for such DSC Committee and Class):
 - (a) if ANC is zero (0), then paragraph 2.2.9 applies;
 - (b) if ANC is one (1), then the single Candidate will be appointed as a Shipper User Representative, and will hold two (2) votes as provided in GT Section D4.3.1;
 - (c) if ANC is two (2), then the two Candidates are appointed as the Shipper User Representatives; or
 - (d) if ANC is greater than two (2), then the two Candidates with the highest number of votes are appointed as the Shipper User Representatives.
- 2.2.9 Where this paragraph 2.2.9 applies in relation to a Class, the following applies for each other Class ("Other Class"):
 - (a) for each (if any) Other Class for which ANC is not less than three (3) the Candidate in such Other Class with the third (3rd) highest number of votes is appointed as an additional Shipper User Representative of that Other Class;
 - (b) for each Other Class for which ANC is less than three (3):
 - where two (2) Shipper User Representatives have been appointed for such Other Class, one of such Shipper User Representatives will hold two (2) votes as provided for in GT Section D4.3.1, and the Code Administrator shall pick the relevant individual at random;
 - (ii) one (1) Shipper User Representative has been appointed, such Shipper User Representatives will hold three (3) votes as provided for in GT Section D4.3.1.
- 2.2.10 Where for the purposes of making an appointment under paragraphs 2.2.8(d) or 2.2.9(a) two (2) or more Candidates hold an equal number of votes ("relevant" Candidates) the Code Administrator shall notify Nominating Shipper Users in the relevant Class of the names of the relevant Candidates and invite each Nominating Shipper User to vote for one of the relevant Candidates (following which vote an appointment will be made in accordance with paragraph 2.2.8(d) or 2.2.9(a) or a further invitation made in accordance with this paragraph 2.2.10).

© 2005 all rights reserved

GTD 45

3 Transporter Representatives

3.1 NTS Representatives

3.1.1 National Gas Transmission shall appoint two (2) representatives to each DSC Committee by notice to the Code Administrator.

3.2 DNO Representatives

- 3.2.1 For the purposes of paragraph 3.2.2:
 - (a) **"Larger DN Operator"** is a DN Operator which operates an LDZ(s) with, in aggregate, four million (4,000,000) or more Supply Points;
 - (b) "Smaller DN Operator" is a DN Operator which is not a Larger DN Operator.
- 3.2.2 The DN Operators collectively shall appoint two (2) representatives to each DSC Committee, one (1) appointed by the Larger DN Operators and one (1) appointed by the Smaller DN Operators; and the DN Operators shall authorise one DN Operator to notify the representatives to the Code Administrator by way of a notice signed by each of the DN Operators.

3.3 IGT Representatives

- 3.3.1 The Independent Networks Association (INA) shall appoint two (2) representatives to each DSC Committee by notice to the Code Administrator.
- 3.3.2 In the event the INA:
 - (a) only appoint one (1) representative, then the individual nominated will be appointed as an IGT Representative, and will hold two (2) votes as provided for in GT Section D4.3.1;
 - (b) appoint no representatives, National Gas Transmission, and the DN Operators collectively, shall each appoint one (1) additional representative.

4 Appointment

- 4.1 National Gas Transmission shall not later than 1 September in Appointment Year Y-1, notify the Committee Secretary of the identity of the individuals to be appointed as the NTS Representatives for Appointment Year Y.
- 4.2 The DN Operators shall not later than 1 September in Appointment Year Y-1, notify the Committee Secretary of the identity of the individuals to be appointed as the DNO Representatives for Appointment Year Y.
- 4.3 The INA shall not later than 1 September in Appointment Year Y-1, notify the Committee Secretary of the identity of the individuals to be appointed as the IGT Representatives for Appointment Year Y.

5 Retirement and ceasing to be a representative

5.1 Each Committee Representative on each DSC Committee shall retire on the 30 September in Appointment Year Y; and a retiring Committee Representative shall be

© 2005 all rights reserved

GTD 46

eligible for re-appointment.

- 5.2 In the event a Shipper User Representative who when nominated was employed or engaged by a Shipper User ceases to be so employed or engaged by the Shipper User, the Shipper User Representative shall be deemed to retire, and it shall be the responsibility of the relevant Shipper User to notify the Committee Secretary of the Shipper User Representative's effective date of retirement.
- 5.3 If any Committee Representative shall for whatever reason (other than retirement pursuant to paragraph 5.1 or 5.2) notify the Committee Secretary that they wish to cease to be a Committee Representative, they shall cease to be a Committee Representative in accordance with the notice.
- 5.4 Where the Committee Secretary is notified under paragraphs 5.2 or 5.3 the Committee Secretary shall inform, in due course, any relevant persons determined by the Committee Secretary.

6 Notices

- 6.1 Each individual who is to be appointed as a Committee Representative shall be entitled to receive notice of any meeting of the relevant DSC Committee which is to take place after the date from which their appointment is effective.
- 6.2 Any Committee Representative who is, pursuant to paragraph 5.1 or 5.2 to retire (and who is not re-appointed) shall not be entitled to receive notice of any meeting of the relevant DSC Committee which is to take place after the date from which their retirement is effective.

7 Vacancies

- 7.1 Subject to paragraph 8.2, where pursuant to paragraphs 5.1, 5.2 or 5.3 or otherwise, a vacancy shall arise, in the case of:
 - (a) a Shipper User Representative, a replacement shall be appointed as a Shipper Users Representative for the relevant Class in accordance with paragraph 2.2;
 - a NTS Representative, National Gas Transmission shall, by notice to the Committee Secretary, identify another individual to be appointed as a NTS Representative;
 - a DNO Representative, the relevant DN Operator shall notify the Committee Secretary of the identity of a replacement to be appointed as a DNO Representative;
 - (d) an IGT Representative, the INA may, by notice to the Committee Secretary, identify another individual to be appointed as an IGT Representative.

8 Alternates

- 8.1 A Committee Representative may only appoint an alternate who is an existing Committee Representative with the agreement of the Committee Chairperson.
- 8.2 Where a Committee Representative with an appointed standing alternate (who is not another Committee Representative) is deemed to retire under paragraph 5.2 or ceases to

© 2005 all rights reserved

GTD 47

be a Committee Representative under paragraph 5.3, such alternate shall be appointed as the replacement Committee Representative until the end of the relevant Appointment Year.

9 Request for a mid-year change of representation

- 9.1 A party may request a change of Committee Representative before 1 July in any Appointment Year where:
 - (a) all the votes attributable to a particular class or party were transferred to another class or party as a result of the previous election cycle (for example, but not limited to, as a result of paragraphs 2.2.9 or 3.3.2(b) above); or
 - (b) there is a party who, part way through the Appointment Year, wishes to fulfil a vacancy within their class of Representatives, where the vacancy is due to insufficient elected Representatives for that class having been appointed in the previous election cycle (for example, but not limited to, as a result of paragraphs 2.2.8(b) and 3.3.2(a)).
- 9.2 Where a party wishes to represent their class in accordance with paragraph 9.1 above, they should submit a joint request for a change of representation to the Code Administrator (in the form requested by the Code Administrator), which is supported by a proposer and a seconder from the same class or party. The request should include a statement of the candidate's willingness to act as a Committee Representative for their class or party.
- 9.3 Following receipt of a request under paragraph 9.2, the Code Administrator shall:
 - notify the Single Point of Contact ("SPOC") for the affected class or party of the request to represent them;
 - (b) notify the existing Committee Representative who may be replaced by the request for a change of representative, for example, a Committee Representative appointed under paragraph 2.2.8(b), 2.2.9 or 3.3.2;
 - (c) notify the relevant class of users or party of the details of the party who has applied for a change of representative and invite any further nominations by a date specified by the Code Administrator (not less than 15 days from the date of the notification). Any further nominations must be supported by a proposer and seconder from the same class or party and include a statement of the nominee's willingness to act.
- 9.4 In the event that there are no further nominations, the Code Administrator shall:
 - notify the party who requested the change of representative of their appointment, which will take immediate effect;
 - (b) notify the existing Committee Representative who is being replaced. This existing Committee Representative will be deemed to have retired with immediate effect following such notification from the Code Administrator; and
 - (c) notify all parties of the change of representation.
- 9.5 In the event that further nominations are received by the Code Administrator under

© 2005 all rights reserved

GTD 48

paragraph 9.3 (c), the Code Administrator shall conduct an election by inviting each member of that party or class to vote by written notice in such form and by such date and time (not less than 5 days) as the Code Administrator may require.

- 9.6 Following any election under paragraph 9, the Code Administrator shall:
 - (a) appoint the nominee with the highest number of votes, such appointment to take immediate effect;
 - (b) notify the existing Committee Representative who is being replaced. This existing Committee Representative will be deemed to have retired with immediate effect following such notification from the Code Administrator; and
 - (c) notify all parties of the change of representation.

© 2005 all rights reserved

GTD 49

© 2005 all rights reserved

GTD 50